Author's response to reviews

Title: MRSA prevalence in European healthcare settings: a review

Authors:

  Madeleine Dulon (madeleine.dulon@bgw-online.de)
  Frank Haamann (frank.haamann@bgw-online.de)
  Claudia Peters (c.peters@uke.uni-hamburg.de)
  Anja Schablon (a.schablon@uke.uni-hamburg.de)
  Albert Nienhaus (a.nienhaus@uke.uni-hamburg.de)

Version: 3 Date: 17 May 2011

Author's response to reviews:

Dear Editorial Team,

we have made the following changes in the manuscript:

Major revisions:
Abstract: The abstract in the manuscript is identical with the version in the submission system.
Acknowledgement: We have included in this section the funding of the study.
References: Reference 6 has been cited with the name of the journal, number of the volume and the pages.
Tables: The shading within the tables has been deleted. We are not able to highlight the title of the columns by making them bold, because otherwise the layout of the tables would be disturbed.

Minor revisions:
Highlighting/tracking: We have deleted all ‘tracked changes’ within the manuscript.
References: Each citation with the term ‘et al.’ has been changed accordingly.
Tables: the visible vertical lines have been removed from the tables.
Typography: the manuscript has been checked for typographical errors.

Sincerely
Dr. M. Dulon